Author's response to reviews

Title: Randomized Pilot Study to Disseminate Caries-Control Services in Dentist Offices

Authors:

David Grembowski (grem@u.washington.edu)
Charles Spiekerman (cspieker@u.washington.edu)
Michael A del Aguila (michael.del_aguila@roche.com)
Maxwell Anderson (maxscruiser@gmail.com)
Debra Reynolds (DReynolds@DeltaDentalWA.com)
Allison Ellersick (AllisonEllersick@yahoo.com)
James Foster (JsFoster@DentalDAC.com)
Leslie Choate (lchoate@everett.wednet.edu)

Version: 3 Date: 27 April 2006

Author's response to reviews:

We have made format changes in the paper to meet the format requirements of BioMed Central. These changes include:

1. Revision of title letters and adding the word "pilot"
2. Change format and content of cover page information
3. Addition of a new Acknowledgements section
4. Format changes to tables

Please note that the Reviewers recommended that the paper's Figure be deleted from the paper, and we have done so in the manuscript. We experienced difficulty in deleting the Figure on the BioMed Central website when we submitted the revised edition of the paper in March.